CrossMark

Citation: Wangchuk P, Pearson MS, Giacomin PR, Becker L, Sotillo J, Pickering D, et al. (2016) Compounds Derived from the Bhutanese Daisy, *Ajania nubigena*, Demonstrate Dual Anthelmintic Activity against *Schistosoma mansoni* and *Trichuris muris*. PLoS Negl Trop Dis 10(8): e0004908. doi:10.1371/journal.pntd.0004908

Editor: Patrick J. Lammie, Centers for Disease Control and Prevention, UNITED STATES

Received: February 2, 2016

Accepted: July 15, 2016

Published: August 4, 2016

Copyright: © 2016 Wangchuk et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files (<u>S1 Fig</u> and <u>S2 Fig</u>).

Funding: This research was funded by a program grant from the National Health and Medical Research Council (NHMRC) awarded to AL, grant number APP1037304. URL: <u>https://www.nhmrc.gov.au/grantsfunding</u>. PW was supported by a NHMRC Peter Doherty Early Career Researcher fellowship number APP1091011. URL: <u>https://www.nhmrc.gov.au/grantsfunding</u>. The funders had no role in study design, RESEARCH ARTICLE

Compounds Derived from the Bhutanese Daisy, *Ajania nubigena*, Demonstrate Dual Anthelmintic Activity against *Schistosoma mansoni* and *Trichuris muris*

Phurpa Wangchuk*, Mark S. Pearson, Paul R. Giacomin, Luke Becker, Javier Sotillo, Darren Pickering, Michael J. Smout[®], Alex Loukas[®]

Centre for Biodiscovery and Molecular Development of Therapeutics, Queensland Tropical Health Alliance, Australian Institute of Tropical Health and Medicine, James Cook University, Cairns Campus, Cairns, Australia

• These authors contributed equally to this work.

* phurpa.wangchuk@jcu.edu.au

Abstract

Background

Whipworms and blood flukes combined infect almost one billion people in developing countries. Only a handful of anthelmintic drugs are currently available to treat these infections effectively; there is therefore an urgent need for new generations of anthelmintic compounds. Medicinal plants have presented as a viable source of new parasiticides. *Ajania nubigena*, the Bhutanese daisy, has been used in Bhutanese traditional medicine for treating various diseases and our previous studies revealed that small molecules from this plant have antimalarial properties. Encouraged by these findings, we screened four major compounds isolated from *A. nubigena* for their anthelmintic properties.

Methodology/Principal Findings

Here we studied four major compounds derived from *A. nubigena* for their anthelmintic properties against the nematode whipworm *Trichuris muris* and the platyhelminth blood fluke *Schistosoma mansoni* using the xWORM assay technique. Of four compounds tested, two compounds—luteolin (**3**) and (3*R*,6*R*)-linalool oxide acetate (**1**)—showed dual anthelmintic activity against *S. mansoni* (IC₅₀ range = 5.8–36.9 µg/mL) and *T. muris* (IC₅₀ range = 9.7–20.4 µg/mL). Using scanning electron microscopy, we determined luteolin as the most efficacious compound against both parasites and additionally was found effective against the schistosomula, the infective stage of *S. mansoni* (IC₅₀ = 13.3 µg/mL). Luteolin induced tegumental damage to *S. mansoni* and affected the cuticle, bacillary bands and bacillary glands of *T. muris*. Our *in vivo* assessment of luteolin (**3**) against *T. muris* infection at a single oral dosing of 100 mg/kg, despite being significantly (27.6%) better than the untreated control group, was markedly weaker than mebendazole (93.1%) in reducing the worm burden in mice.

data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Conclusions/Significance

Among the four compounds tested, luteolin demonstrated the best broad-spectrum activity against two different helminths—*T. muris* and *S. mansoni*—and was effective against juvenile schistosomes, the stage that is refractory to the current gold standard drug, praziquantel. Medicinal chemistry optimisation including cytotoxicity analysis, analogue development and structure-activity relationship studies are warranted and could lead to the identification of more potent chemical entities for the control of parasitic helminths of humans and animals.

Author Summary

Schistosomiasis and trichuriasis affects millions of people worldwide and are caused by blood flukes and whipworms, respectively. Only a handful of anthelmintic drugs exist to treat these infections and the pipeline for the next generation of anthelmintic drugs is sparse, precipitating the need for new drug development. In this context, medicinal plants present a viable source of novel anthelmintic compounds. This inspired us to study the selected naturally occurring compounds derived from a Bhutanese daisy medicinal plant, Ajania nubigena for their anthelmintic activities. Here, using the xWORM motility assay, we demonstrate that two compounds, luteolin (3) and (3R,6R)-linalool oxide acetate (1), display significant broad-spectrum anthelmintic activity against two of the most important genera of human helminth parasites, the nematode whipworm and the platyhelminth blood fluke. Luteolin exhibited the best activities with IC₅₀ values of 5.8 µg/mL against schistosomes and 9.7 µg/mL against whipworms. Using scanning electron microscopy we showed that luteolin damages the tegument of blood flukes and induces abnormalities in the bacillary bands/glands and cuticles of whipworms. Intriguingly, our previous study showed that luteolin (3) was effective against multi-drug resistant Plasmodium falciparum malaria. Due to its broad-spectrum anti-parasitic activities, luteolin (3) is a desirable drug lead scaffold, which could be used for developing effective compounds to control and treat numerous tropical diseases.

Introduction

The World Health Organization (WHO) recognises 17 different 'neglected tropical diseases' (NTDs) that affect more than 1.4 billion people in 149 countries [1]. Helminth infections caused by roundworms (nematodes) and flatworms (platyhelminths) comprise the largest group of NTDs [2]. Whipworms (Nematoda) cause trichuriasis and infect about 800 million people worldwide, second among the nematodes only to Ascaris infection [3]. The schistosome blood flukes (Platyhelminthes) cause schistosomiasis, a disease that afflicts more than 240 million individuals and kills hundreds of thousands each year [4].

A variety of approaches have been employed to combat these infections including education, vector control, sanitation and hygiene, behavioural change and mass drug administration (MDA) programs. Various *in vitro* and animal model studies have highlighted the repurposing of existing drugs and discovery and development efforts for new drugs [2, 5] but all things considered, the pipeline for the next generation of anthelmintic drugs is sparse. Indeed, a systematic assessment of databases of drug regulatory authorities and the WHO, as well as clinical trial registries, revealed that no new antiparasitic drugs have been approved during the last decade [6]. There are only a handful of anthelmintic drugs on the market, some of which have unwanted side effects or achieve poor cure rates due to primary drug resistance developing in the parasites [7-9].

For example, praziquantel, which is the sole frontline drug used in the mass treatment of schistosomiasis, is efficacious but has many disadvantages: a) it is ineffective against juvenile stages of the parasite, b) reduced efficacy has been reported in field studies [10], c) there is a strong possibility that praziquantel resistance could appear if sufficient selection pressure is applied and mass drug administration is continued [11], and d) its active (*S*)-enantiomer and inactive (*R*)-enantiomer components remain inseparable in the production process, rendering bulky tablets that discourages patients from taking the right doses or the complete dosing regimen, which could trigger the development of drug resistance [12]. Until new arsenals of safe and effective drugs and/or vaccines are made available, helminth infections will continue to affect the world's most impoverished populations, causing significant morbidity and mortality worldwide.

While new drugs can be developed synthetically, natural products-especially the medicinal plants—have been an important pool of antiparasitic drugs. Quinine and artemisinin discovered from medicinal plants continue to save the lives of millions of people worldwide. As such, the notion of therapeutics derived from medicinal plants has re-surfaced [13]. Crude extracts and compounds of plant origin have been demonstrated to possess broad biological activities in *in vitro* and *ex vivo* assays and animal models of parasitic infections [14-19]. Edwards *et al.* [20] showed that 7-keto-sempervirol isolated from the boxthorn from which goji berries are harvested, Lycium chinense, was effective against Schistosoma mansoni and Fasciola hepatica. A compound that displays such broad anti-parasitic activity against various life stages of multiple parasites is highly desirable. Extracts of the Bhutanese medicinal plant from the flowering daisy family, Ajania nubigena (Syn. Tanacetum nubigenum DC.) have been previously shown to possess broad biological activities including antiparasitic effects against Plasmodium falciparum and antimicrobial properties [21]. It is locally known as *m.khan-d.kar* and has been used in Bhutanese traditional medicine (derived from Tibetan scholarly medicine) for thousands of years as incense and for treating an array of conditions and infections including wounds, bleeding and swelling [21]. Although this plant is not specifically indicated for treating intestinal worms, the decoction of its closely related species, Tanacetum parthenium L. (feverfew) and Tanacetum dolichophyllum Kitam has been traditionally used by the Ladakhis Amchis (medical system derived from Tibetan medicine and similar to Bhutanese traditional medicine) [22] and Costa Ricans healers [23] against intestinal worms. These plants have reserves of highly aromatic essentials oils that have evolved to aid in plant protection and competition against plant parasites and herbivorous insects. Chemically, these plants contain similar chemotypes including sesquiterpenes and flavonoids [23]. Encouraged by these lead information, we have investigated the anthelmintic properties of four compounds isolated from the Bhutanese A. nubigena against two of the most important genera of human helminth parasites, the nematode whipworm (Trichuris) and the platyhelminth blood fluke (Schistosoma). To monitor worm viability we used xWORM, a technique that monitors helminth motility in real time using xCELLigence [24–25]. The advantage of using xWORM over other methods is that it enables high-throughput screening of a large number of compounds in a fully automated, label-free manner.

Materials and Methods

Plant materials and preparation of compounds

The aerial part of wild *Ajania nubigena* was collected from alpine mountains (altitude range of 3600–4800 meters above sea level) of Lingzhi, Bhutan in August 2009. The collected plant

material was air-dried and a herbarium specimen with voucher number 73 was deposited at the herbarium collection section of Menjong Sorig Pharmaceuticals, Ministry of Health, Bhutan. The air-dried plant material (2 kg) was chopped into small pieces and was repeatedly extracted with methanol (AR/HPLC grade, 3 L over 48 h). The extract was filtered and then concentrated using a Buchi rotary evaporator to generate a crude methanol (MeOH) extract (58.2 g). The isolation technique was described previously [21]. MeOH extract was dissolved in MeOH:H₂O (200 mL, 1:9) and then fractionated with hexane followed by ethyl acetate to obtain the hexane extract (28.0 g) and the ethyl acetate extract (12.5 g), respectively. Subsequently, essential oil (EO) extraction was performed using hydro-distillation (60°C). One kg of dried plant material yielded 7 mL of pale green EO. The crude MeOH extract and EO were subjected to extensive natural products isolation processes. Flash column chromatography packed with Merck Kieselgel 60 PF254 and pre-coated silica plates (0.2 mm silica thickness, Merck) were used for repeated separation and purification of compounds. UV light (short wavelength of 254 nm, long wavelength of 366 nm) and ceric ammonium molybdate (CAM) were used for visualization and detection of compounds on Thin Layer Chromatography (TLC) plates. Eight compounds were isolated and characterised in total from the MeOH and EO extracts using Infrared (IR) Spectroscopy, Mass Spectrometry (ESI-MS, HR-EI-MS), Gas Chromatography Mass Spectrometry (GCMS), and Nuclear Magnetic Resonance (NMR-¹H, ¹³C, gCOSY, gNOESY, TOCSY, gHSQC and gHMBC) [21]. In this study, we have selected four major compounds whose structures are produced in Fig 1: (3R,6R)-linalool oxide acetate (1), (E)-spiroether (2), luteolin (3) and luteolin-7-O- β -D-glucopyranoside (4).

The stock solutions of the four test compounds were prepared at the concentration of 100 mg/mL in DMSO and then subsequently diluted them with respective tissue culture media to make 10x solutions. 20 μ l of 10× drug concentration was added to 180 μ l of media containing helminths in the E-plate wells. Control worms were cultured in the presence of DMSO equivalent to that used for the highest drug concentration; this group was used to determine 100% motility. For schistosomula drug assays, stock solutions were diluted in culture media with two-fold dilutions with in-well concentrations of (2–1000 μ g/mL).



Fig 1. Structure of compounds [21]. (3*R*,6*R*)-linalool oxide acetate (1), (*E*)-spiroether (2), luteolin (3), luteolin-7-*O*-β-D-glucopyranoside (4) were derived from *A. nubigena* and tested for their anthelmintic activities using xWORM. Plant photo courtesy–PW.

doi:10.1371/journal.pntd.0004908.g001

Preparation of Schistosoma mansoni

S. mansoni cercariae were shed from infected *Biomphalaria glabrata* snails (Biomedical Research Institute, MD, USA) by exposure to light at 26°C for 2 hours and used to infect 12–14 week old male BALB/c mice (120 cercariae/mouse) by abdominal penetration [26]. Adult flukes were perfused from the mesenteries 7 weeks post-infection and then transferred to Basch medium for culturing as previously reported [27]. To monitor the effects of drugs on *S. mansoni* schistosomula, shedding of cercariae from snails and subsequent *in vitro* transformation to schistosomula was performed as described by Peak et al [28].

Preparation of Trichuris muris

Genetically susceptible mice (STAT6⁻¹⁻) were orally infected with *T. muris* eggs (200 μ L volume containing ~ 200 eggs) and sacrificed after 4 weeks. Adult worms were harvested from the caecum, washed with PBS/2x antibiotic/antimycotic (AA) and resuspended in 100 μ l of RPMI containing 10% foetal calf serum and AA (culture medium) then transferred to E-plates for motility assessment using the xWORM assay.

xWORM assay

The trematocidal effects of test compounds against *S. mansoni* were evaluated using an xCEL-Ligence SP system (ACEA Biosciences) as described by us [25]. Adult flukes (1 fluke per well) were placed in triplicate for each compound into 96 well E-plates (ACEA Biosciences) containing 180 μ l of culture medium and cultured overnight at 37°C with 5% CO₂ to obtain a baseline motility reading. Test compounds were added to E-plates and motility was monitored for 12–40 hr. The inter-well spaces of E-plates were filled with 100 μ L culture media. All experiments were carried out as per manufacturer's instructions with 15 sec read intervals using the real time cell assay (RTCA) software (ACEA Biosciences) as described previously [24–25].

Similarly, the nematocidal effects of test compounds against *T. muris* were assessed using the same xCELLigence SP system as described above. We determined the optimal culture duration and worm concentration to maximize the signal to noise ratio using the xWORM technique for the first time with *T. muris*. Different numbers of adult *T. muris* (2, 4 and 8) were added to individual E-plate wells and motility was monitored overnight. Four worms of mixed gender per well in a final volume of 200 μ l of culture medium was determined to be optimal for this study. The E-plates containing worms were treated with prepared concentrations of the test compounds and were monitored using the xCELLigence SP system for 12–40 hr. Inter-well spaces were filled with 100 μ L of culture medium or PBS to prevent evaporation. Each set of conditions was monitored in triplicate.

Determining the effects of compounds against schistosomula

The 96 well plates containing culture media were loaded with schistosomula (100 μ L volume containing ~ 100 schistosomula) in triplicate and treated with the test compounds at various in-well concentrations of 2–1000 μ g/mL. Plates were cultured at 37°C with 5% CO₂ for 12–40 hr and were finally stained with trypan blue solution to assess final viability after treatment. The stained schistosomula were observed by light microscopy, and live and dead flukes in each well were counted manually and 50% inhibitory concentration (IC₅₀) values were obtained.

Determination of IC₅₀ values for test compounds

The IC_{50} values of test compounds were determined based on the motility index for adult worms as described by us [25]. Briefly, motility index was calculated as the standard deviation

(SD) over 800 data points (i.e. 4 readings per min for 200 min) of the cell index (CI) difference from the rolling average over 20 data points (10 proceeding and preceding CI values—5 min total). One hundred percent motility was determined from the average motility index of the untreated wells, while 0% motility was determined from a media only well (no worms present). The motility index averaged over 100 data points (25 min) was converted to percentage motility and this figure was used in GraphPad Prism 6.0 to calculate dose response curves. We used a log (test compound concentration) vs normalised response (100%–0%) formula, with variable slope when data were sufficient or set -1 hill slope when data was limited, and automatic removal of outliers (with default ROUT coefficient used: Q = 1.0%). IC₅₀ values for each dose concentration were calculated at 1 hr, 6 hr, and 12 hr post-treatment of the worms with the test compounds. Compounds with IC₅₀ values of higher than 100 µg/mL were considered ineffective in this study.

Statistical analysis

Statistical analyses were undertaken using GraphPad Prism 6.0. When data were sufficient to use the variable slope analysis, the Hill Slope and the Log IC_{50} value were together compared for significant differences using an extra sum-of squares F-test. For the *in vivo* mouse experiments, 1-way ANOVA with Holm-Sidak's multiple comparisons test was used for determining significance *p*-values.

Scanning electron microscopy

Worms treated with the test compounds were prepared for scanning electron microcopy (SEM) as follows: a) fixed in 3% gluteraldehyde in Sorensen's buffer overnight, b) dehydrated for 15 min in a graded ethanol series (50%, 60%, 70%, 80%, 90%, 100%), mixture of ethanol and hexamethyldisilizane (HMDS) (1:1 ratio) and then finally with pure HMDS (100%), c) the dehydrated worms were covered and left overnight in a fume hood to allow the HMDS to evaporate. Completely dried worms were placed on an aluminum stub (at least three worms from each treatment regimen), sputtered with gold and visualized using a JEOL JSM scanning electron microscope operating at 10 kV. Each worm on a stub was scanned from head to tail to determine if the compounds had altered its gross morphology. Digital image acquisition was performed on the affected region of the worms using Semaphore software.

In vivo animal model studies of luteolin (3)

Four to five week-old STAT6⁻¹⁻ mice were grouped (each group with 9 mice) as: solvent control, positive control and luteolin (3). Each mouse was orally infected with 200 μ l of PBS containing approximately 200 live embyronated eggs of *T. muris*. These mice were housed for 4 weeks with constant access to water and pelleted food. After 4 weeks, luteolin (3) and the positive control drug (mebendazole) prepared in 1% DMSO/PBS were orally administered at a single dose of 100 mg/kg (9 mice in total for each group) as per the protocol [29–30]. Five days after one dose of oral treatment the mice were sacrificed, worms were harvested from the caecum, and counted manually using light microscopy. The recorded numbers of worms were averaged to find the percentage reduction in worm burden for each group of mice.

Ethics and approvals

The permit to collect medicinal plants from the park management areas around Lingzhi, Bhutan was obtained from the Department of Forest, Ministry of Agriculture and Forestry in Bhutan. The material transfer agreement and approval was sought from the National Biodiversity Centre of Bhutan. MeOH extracts of the plants were transported to Australia with prior approval from the Bhutan Agriculture and Food Regulatory Authority, University of Wollongong and sample inspections by Australian Quarantine & Inspection Service in 2010. The James Cook University (JCU) animal ethics committee approved all experimental work involving animals (Ethic approval number A2213). Mice infected with *S. mansoni* and *T. muris* were raised in cages in the JCU animal facility for 4–7 weeks in compliance with the Australian Code of Practice for the Care and Use of Animals for Scientific Purposes, 7th edition, 2007 and the Queensland Animal Care and Protection Act 2001. Mice were kept under normal conditions at regulated temperature (22°C) and lighting (12 hr light/dark cycle) with free access to pelleted food and water. All reasonable efforts were made to minimise the suffering of the mice.

Results

The isolation and characterization of compounds from the *A. nubigena* plant were performed as reported previously by us [21]. Out of eight compounds isolated from this plant, several were reported to have antimalarial, antibacterial, antifungal and cytotoxicity activities [21]. Encouraged by the antimalarial activities, four major compounds (Fig 1): (3*R*,6*R*)-linalool oxide acetate (1), (*E*)-spiroether (2), luteolin (3) and luteolin-7-*O*- β -D-glucopyranoside (4), were tested for broad-spectrum anthelmintic activities against two distinct phyla of helminths, the platyhelminth trematode blood fluke *S. mansoni* and the nematode whipworm *T. muris*.

Trematocidal activity of compounds against adult S. mansoni

Of the four compounds tested, (3R,6R)-linalool oxide acetate (1), luteolin (3) and luteolin-7-O- β -D-glucopyranoside (4) showed anti-schistosome dose-dependent anti-schistosome effects (Fig 2A). At the highest concentration tested (1 mg/mL) all three compounds killed flukes within 1–12 hr. Lower drug concentrations, however, took longer to kill flukes reflected by higher motility index values (Fig 2A).

When the IC₅₀ values of each compound were averaged or combined (calculated for the dose concentrations of 0.1–1000 µg/mL) for each time point (1 hr, 6 hr and 12 hr), luteolin (3) and luteolin-7-O- β -D-glucopyranoside (4) appeared to be fast acting on worms as their IC₅₀ values did not change significantly between the initial and final 12 hr dosing time points (Fig 2B). On the other hand, (3*R*,6*R*)-linalool oxide acetate (1) was slow-acting, showing two-fold decreases in IC₅₀ at each 6 hr time point. Of the compounds assessed, luteolin (3) exhibited significantly better trematocidal activity at both 6 hr and 12 hr time points with IC₅₀ values of 4.6 µg/mL and 5.8 µg/mL, respectively.

Effects of compounds against schistosomula of S. mansoni

The compounds that were active against adult *S. mansoni* were also tested against the intramammalian larval stage, the schistosomulum. Monitoring of schistosomula survival in the presence of different drug concentration using Trypan blue exclusion showed that luteolin (**3**) started to show lethal effects at the lowest dilution of $3.91 \mu g/mL$ and achieved 98-100% killing at the dilution of $31.3 \mu g/mL$. (3R,6R)-linalool oxide acetate (**1**) however started to show lethal effects at a concentration of $125 \mu g/mL$ and only achieved about 43% killing at the maximum dose tested of $250 \mu g/mL$. Schistosomula treated with 1% DMSO alone (solvent control) had 100% survival as measured by Trypan blue exclusion. A dose response curve of schistosomula survival after treatment with luteolin (**3**) revealed an IC₅₀ of $13.3 \mu g/ml$ (Fig <u>3B</u>).

Effects of luteolin (3) on the morphology of adult S. mansoni

Luteolin (3) demonstrated the best anthelmintic activity when motility was assessed using xWORM. The effect of this compound and praziquantel on adult fluke morphology at



Fig 2. *In vitro* anti-schistosome activities of four compounds against adult *S. mansoni*. A) Motility index dose response curve of worms at the 12 hr time point when treated with (3R,6R)-linalool oxide acetate (1), (E)-spiroether (2), luteolin (3), luteolin-7-O-β-D-glucopyranoside (4) (abbreviated as Com-1, Com-2, Com-3 and Com-4) at different concentrations (0.1–1000 µg/mL). Motility index was calculated as the standard deviation (SD) over 800 data points (i.e. 4 readings per min for 200 min) of the cell index (CI) difference from the rolling average over 20 data points (10 proceeding and preceding CI values—5 min total). B) 50% inhibitory concentration (IC₅₀) curves over time. Error bars represent 95% confidence intervals of nonlinear curve fit. The curves were marginally shifted on the x-axis to aid viewing. These figures represent the data from three independent studies.

 $4-20 \ \mu\text{g/mL}$ concentration, with particular emphasis on the tegument, was assessed by SEM. Adult flukes treated with praziquantel adopted a tightly coiled appearance due to contraction. Both male and female flukes treated with luteolin (3) were contracted and coiled compared to control flukes cultured in 1% DMSO in media, but not as tightly coiled as praziquantel-treated parasites (Fig 4A-4F).

Observation of fluke teguments by SEM under high magnification revealed that the tegument of DMSO-treated flukes displayed numerous tubercles, with well-formed spines in the males (Fig 5A) and clearly defined surface grooves with sensory papillae in females (Fig 5B), and oral suckers with clearly defined pits containing sharp spines (Fig 5C). Flukes treated with praziquantel (Fig 5D–5F) and luteolin (3) (Fig 5G–5I) exhibited severe morphological alterations of the tegument.

At the lowest concentration tested of 4 μ g/mL we observed clear tegumental damage induced by luteolin (3) (Fig 5G–5I), similar to that induced by praziquantel (Fig 5D–5F). While male flukes (Fig 5G) suffered partial loss of pits and their encased spines, female flukes (Fig 5E) showed surface erosion, constriction of grooves, bursting of small sensory papillae and formation of holes on the tegument. At higher concentrations of 20–1000 μ g/mL, luteolin (3) and praziquantel completely destroyed the body surfaces and exhibited erosion of tubercles, oral and ventral suckers (S1 Fig).



Fig 3. Effect of compounds on the survival of schistosomula. A) Images (20x) of wells containing schistosomula that were treated with linalool oxide acetate (1) and luteolin (3) (abbreviated as Com-1 and Com-3) at the maximum concentration tested of 250 μ g/mL and stained with trypan blue. Deep blue staining signifies a dead parasite. B) The effect of (3*R*,6*R*)-linalool oxide acetate (1) and luteolin (3) (abbreviated as Com-1 and Com-3) on schistosomula mortality observed at two-fold dilutions (250–1.9 μ g/mL). The data was generated from triplicate samples obtained from two independent studies.

Nematocidal activities of compounds against adult T. muris

Prior to testing the nematocidal effects of (3R,6R)-linalool oxide acetate (1), (*E*)-spiroether (2), luteolin (3) and luteolin-7-*O*- β -D-glucopyranoside (4), we standardized the culturing conditions of adult *T. muris* for the xWORM assay. E-plate wells containing four adult worms (both males and females) yielded optimal motility signals (S2 Fig), and was the condition selected for subsequent anthelmintic screening of the test compounds. (3*R*,6*R*)-linalool oxide acetate (1) and luteolin (3) showed the best anti-*Trichuris* activity with IC₅₀ values of 20.4 µg/mL and 9.7 µg/mL, respectively, calculated on cell motility index at the 12 hr time point (Fig 6A).

The IC₅₀ values of the four compounds tested here were obtained using xWORM and calculated at 1 hr, 6 hr and 12 hr time points (Fig 6B). Luteolin (3) was the most efficacious drug in terms of reduced motility of *T. muris*, exhibiting the lowest or equally low IC₅₀ values at all time points and a final 12 hr value of 9.7 μ g/mL.

Effects of luteolin (3) on the morphology of adult T. muris

Based on the efficacy of luteolin (3) at reducing *Trichuris* motility, we examined the morphological changes in the cuticle induced by this compound using the SEM protocols described by Stepek et al [<u>30</u>] and Tritten et al [<u>31</u>] specific to *T. muris*. Live adult *T. muris* (mixed sexes) were treated for 48 hr with luteolin (3) or mebendazole at dose concentrations of 200–1000 μ g/mL. Morphological changes were observed towards the anterior end of the worms in the form of partially damaged bacillary band/glands and adjacent cuticle. The bacillary band is a



Fig 4. Effects of luteolin (3) and praziquantel on the morphology of adult S. *mansoni* at 4–20 µg/mL concentration visualized using scanning electron microscopy. A, C and E show male flukes. B, D and F depict female flukes. (A, B) Flukes in DMSO/media control. (C, D) Flukes treated with praziquantel (tightly coiled). (E, F) Flukes treated with luteolin (3) (moderately coiled). Three separate wells (with 3–5 adult flukes per well) were treated with different doses of test samples as above and each of them was examined under SEM.

specialized row of longitudinal cells of some nematodes consisting of glandular and non-glandular cells. These bands host the glands. Worms treated with DMSO/media (solvent control) alone had a moderately coiled appearance with a smooth cuticle displaying knitted parallel segmental joins and minimal shrinkage of bands (Fig 7A–7C) in comparison to the groups treated with mebendazole (Fig 7D–7F) and luteolin (3) (Fig 7G–7I). At higher magnification we observed that the luteolin-treated worms exhibited blister-like formations on the surface of the cuticle, swelling and loosening of cuticle seams/grooves near the bacillary glands (Fig 7I). These morphological changes were similar to that of the mebendazole-treated worms (Fig 7F).



Fig 5. Scanning electron micrographs of S. mansoni adult fluke teguments after treatment with 4 μg/mL concentration of luteolin (3) and praziquantel. A, B and C represent flukes treated with solvent (1% DMSO in culture media). A) Male tegument with well defined tubercles. B) Female tegument with clearly defined grooves and sensory papillae. C) Oral sucker with well formed pits containing spines. D, E and F represent flukes treated with praziquantel. D) Male tegument with damaged tubercles and spines. E) Female tegument with damaged papillae and swollen grooves. F) Oral suckers with eroded pits and detached spines. G, H and I represent flukes treated with luteolin (3). G) Male tegument with destroyed tubercles and holes. H) Female tegument with burst sensory papillae and holes. I) Oral suckers with partially eroded pits. Arrows in panels highlight treatment effects described in results text. All images at ×4000 magnification and 1 μm are scale bars shown.

Anti-Trichuris effect of luteolin (3) in mice infected with T. muris

Based on the significant *in vitro* nematocidal activity demonstrated by luteolin (**3**), this compound was further assessed for its anti-*Trichuris* effect *in vivo* using a mouse model of *T. muris* infection [29–31]. Four weeks post-infection with *T. muris* eggs, mice were administered with a single 100 mg/kg oral dose of luteolin (**3**). When mice were sacrificed 5 days later we observed that a single treatment of this compound resulted in a 27.6% reduction in worm burdens (249 worms); luteolin (**3**)–treated mice = 651 worms; DMSO-treated mice = 900 worms; P = 0.0087) (Fig 8). The positive control drug, mebendazole, reduced worm burdens by 93.1% (mebendazole treated mice = 50 worms). The activity of luteolin (**3**) against *T. muris*, despite being significantly better than the untreated control group, was markedly weaker than mebendazole in the mouse model. Mice treated with luteolin (**3**) did not show signs of toxicity at any of the concentrations tested.

Discussion

Globally, helminth infections caused by roundworms (nematodes) and flatworms (platyhelminths) comprise the largest group of NTDs [2]. Schistosomiasis (caused by the platyhelminth blood flukes) and trichuriasis (caused by the nematode whipworms) affect about 240 million and 800 million people, respectively [3–4]. While schistosomiasis is transmitted through infected snails and water, trichuriasis is transmitted through soil and is therefore referred to as a soil-transmitted helminth infections (STHI). The STHI are among the most common



Fig 6. Anti-*Trichuris* activity of compounds (1–4) on adult *T. muris*. A) Motility index dose response curve of worms at the 12 hr time point when treated with (3R,6R)-linalool oxide acetate (1), (*E*)-spiroether (2), luteolin (3) and luteolin-7-*O*- β -D-glucopyranoside (4) (abbreviated as Com-1, Com-2, Com-3 and Com-4) at different concentrations. Motility index was calculated as the standard deviation (SD) over 800 data points (i.e. 4 readings per min for 200 min) of the cell index (CI) difference from the rolling average over 20 data points (10 proceeding and preceding CI values—5 min total). B) Combined IC₅₀ values of these four compounds calculated for three different doses at 1 hr, 6 hr, and 12 hr time points. Error bars represent 95% confidence intervals of nonlinear curve fit. The curves were marginally shifted on the x-axis to aid viewing. These figures represent the data from three independent studies.

infections worldwide and affect the poorest and most deprived communities where sanitation is poor. Sole reliance on praziquantel for schistosomiasis, and only a very small number of drugs (some with poor cure rates) for STHI, has precipitated the need for new anthelmintic drugs to treat parasites that infect both humans and animals [7]. In this context, medicinal plants present a viable source of novel anthelmintic compounds. Indeed, many anti-parasitic drugs including quinine, chloroquine, artemisinin and atovaquone were originally purified from plants. Our initial study, involving the crude CHCl₃ extract of *A. nubigena* (syn. *Tanacetum nubigenum* DC.) and its compounds luteolin and luteolin-7-*O*- β -D-glucopyranoside, showed significant antiparasitic activities against the multidrug resistant K1CB1 strain and chloroquine sensitive TM4/8.2 strain of *Plasmodium falciparum* [16, 21]. Interestingly, this plant and its close relatives have been used in the ethnomedicines for treating arrays of disorders including wound, bleeding and worm infection [16, 21–23].

In this study, we have demonstrated that compounds linalool oxide acetate (1) and luteolin (3) had significant trematocidal activity against *S. mansoni* and nematocidal activity against *T. muris*. These compounds are simple small secondary plant metabolites belonging to the terpenes and flavonoids. Luteolin (3) was the most active compound against both parasites with IC_{50} values of 5.8 µg/mL against *S. mansoni* and 9.7 µg/mL against *T. muris* calculated at the 12 hr time point. It also effectively killed schistosomula with an IC_{50} value of 13.3 µg/mL. Intriguingly, this same compound has been shown to have significant anti-malarial activity against *P. falciparum* TM4/8.2 (chloroquine-antifolate sensitive strain) and K1CB1 (multidrug resistant



Fig 7. Effects of luteolin (3) and mebendazole on the cuticle of adult *T. muris.* A) Normal appearance of control worms cultured in 1% DMSO/media (x100) and at higher magnification focusing on the cuticle (B) and bacillary gland (C), respectively. D) Shrinkage of the anterior regions of worms after treatment with mebendazole and damage to the cuticle (E) and bacillary glands (F) were observed at higher magnification. G) Shrinkage of the anterior regions of worms after treatment with luteolin (3) and damage to the cuticle (H) and swelling of bacillary glands (I) observed at higher magnification. Red arrows in all panels highlight treatment damage described in results text.

strain) [21]. New anti-parasitic drugs require excellent safety and therapeutic profiles, should exhibit broad spectrum activity against different types of infections, and also display significant activity against different developmental stages of parasites. Current anthelmintic drugs are generally effective at treating single stages of target helminths. For example, praziquantel is effective against adult stage schistosomes but not schistosomula/cercariae. Therefore, finding a broad-spectrum drug that could treat multiple diseases or multiple life stages is desirable when treating large populations in resource-poor settings. Luteolin (3) met these criteria in that it has anti-malarial [21], anti-fluke and anti-whipworm properties. In addition, this compound was effective in killing schistosomula, the stage of S. *mansoni* that is unaffected by praziquantel.

While our findings do not specifically address the mechanism of action of these anthelmintic compounds, we showed that luteolin (3) is capable of damaging the outer surface membranes of the parasites-the fluke tegument and the nematode cuticle and associated glandular structures, and worms adopted a coiled state of paralysis. Previous studies on *S. mansoni* have demonstrated that the tegument plays essential roles in many processes at the host-parasite interface, and numerous molecular pathways that are represented at the host-parasite boundary are anti-parasitic drug targets [5, 32]. SEM has been used to demonstrate the mechanisms of many anti-parasitic drugs, including oxamniquine, praziquantel, mefloquine, mebendazole and artemisinin [20, 32-39]. Schistosomes treated with these drugs displayed vacuolization or bubble-like-lesions, surface erosion, destruction of tubercles and tissues, loss of sensory



Fig 8. Effects of luteolin (3) and mebendazole on *T. muris* **burden in mice.** The graph represents the percentage of worms recovered from the STAT6⁻¹⁻ mice five days after a single oral dose of luteolin (3) (27.1% reduction with *p*-value of 0.0087) and the positive control drug, mebendazole (93.1% reduction with *p*-value of 0.0005). The *p*-values were determined by one-way ANOVA Holm-Sidak's multiple comparisons test. Each experimental group consisted of nine mice.

papillae, and pore formation leading to death. SEM has also been used to reveal cuticular damage in nematodes, particularly for the benzimidazole class of drugs [40-41]. Stepek et al [30] and Tritten et al [31] first used the plant extracts and nitazoxanide to demonstrate the mechanisms of the antiparasitic action against *T. muris*. These same morphological changes were observed in adult *S. mansoni* and *T. muris* that were treated with luteolin (3), suggesting that the mechanism of action is similar to at least some of the existing anthelmintic drugs. For *T. muris*, luteolin (3) and mebendazole mainly affected the anterior bacillary band and surrounding glands. The anterior ventrolateral face of the worm contain the glandular pores, bacillary band sensory cells and glands, and the stichosome cells which helps in the formation of the syncytial feeding site in the host, and plays an important role in the excretion of digestive enzymes, pre-digestion and nutrient uptake in *Trichuris* [42-45].

Luteolin (3), with a Log*P* value of 2.6, meets the Lipinski rule of 5 criterion for drug-likeness [46]. Generally, compounds with Log*P* values in the range of 2–3 are more likely to diffuse/permeate through the cell membrane of an organism, and therefore enabling them to interact with target receptors. There was no structural similarity between our active compounds and the currently used anthelmintic drugs, which suggest that compounds that damage the worm surface do not necessarily have similar structural scaffolds. It should be noted that our *in vivo* assessment of luteolin (3) against *T. muris* infection at a single oral dosing of 100 mg/kg, despite being significantly (27.6%) better than the untreated control group, was markedly weaker than mebendazole (93.1%) in reducing the worm burden in mice. Moreover, the mice showed no signs of ill health, suggesting that luteolin (3) is not overtly toxic and allows future studies to explore the efficacy of multiple treatments. We did not assess *in vivo* efficacy of luteolin (3) against *S. mansoni* in the mouse model. Future work will entail synthesis of luteolin and its derivatised compounds in a thorough *in vivo* assessment of efficacy in mouse models of schistosomiasis and trichuriasis, as well as other soil-transmitted helminth infections.

Supporting Information

S1 Fig. SEM images of damaged tubercles, oral and ventral suckers (anterior parts) of *S. mansoni* (male) when treated with luteolin (3) and praziquantel at a dose concentrations of **20–1000 μg/mL**. Praziquantel treated groups (A and C). Luteolin treated groups (C and D). (TIFF)

S2 Fig. Motility signals of *T. muris* (mixed male and female) obtained by xCELLigence with 2–16 worms per well. A) depicts the Motility Index of 2–16 worms per well over 12 hours. B) depicts the average blanked Motility Index of 2–16 worms per well over 12 hours with standard deviation error bars (generated from data shown in A). Four worms per well was determined as optimal as the motility index signal was the most stable over time, with lowest variation. (TIF)

Acknowledgments

We appreciate Dr. Jen Whan of Advanced Analytical Centre (AAC), James Cook University for helping us with the SEM analysis. Mr. Samten of Menjong Sorig Pharmacueitcals, Ministry of Health, Bhutan assisted us with the plant materials. Schistosome infected snails were generously provided by the Biomedical Research Institute (BRI) in USA—a nonprofit 501 c3 company committed to improving global health through a variety of research and service activities.

Author Contributions

Conceived and designed the experiments: PW MJS AL. Performed the experiments: PW MJS MSP PRG DP LB. Analyzed the data: PW MJS AL. Contributed reagents/materials/analysis tools: JS. Wrote the paper: PW MJS JS AL. Obtained the medicinal plants from Bhutan with prior material transfer agreement signed with National Biodiversity Centre of Bhutan: PW.

References

 WHO. Neglected tropical diseases (NTDs). 2015a. Available: <u>http://www.who.int/neglected_diseases/ diseases/en/23/2/15</u>.

- Panic G, Duthaler U, Speich B, Keiser J. Repurposing drugs for the treatment and control of helminth infections. Int J Parasitol Drugs Drug Resist. 2014; 4: 185–200. doi: <u>10.1016/j.ijpddr.2014.07.002</u> PMID: <u>25516827</u>
- USAID. Whipworm (Trichuris trichiura or Trichocephalus trichuris). 2014. Available: http://www.neglecteddiseases.gov/target_diseases/soil_transmitted_helminthiasis/whipworm/index.html/5/3/15.
- WHO. Schistosomiasis: A major public health problem. Available: <u>http://www.who.int/schistosomiasis/en/23/2/15</u>.
- Morel M, Vanderstraete M, Cailliau K, Lescuyer A, Lancelot J, Dissous C. Compound library screening identified Akt/PKB kinase pathway inhibitors as potential key molecules for the development of new chemotherapeutics against schistosomiasis. Int J Parasitol Drugs Drug Resist. 2014; 4: 256–266. doi: <u>10.1016/j.ijpddr.2014.09.004</u> PMID: <u>25516836</u>
- Pedrique B, Strub-Wourgaft N, Some C, Olliaro P, Trouiller P, Ford N, Pécoul B, Bradol JH. The drug and vaccine landscape for neglected diseases (2000–11): a systematic assessment. The Lancet Global Health. 2013; 1: e371–e379. doi: <u>10.1016/S2214-109X(13)70078-0</u> PMID: <u>25104602</u>
- Keiser J, Utzinger J. Efficacy of current drugs against soil-transmitted helminth infections: systematic review and meta-analysis. JAMA. 2008; 299: 1937–1948. doi: <u>10.1001/jama.299.16.1937</u> PMID: <u>18430913</u>
- Wolstenholme AJ, Martin RJ. Anthelmintics-From discovery to resistance. Int J Parasitol Drugs Drug Resist. 2014; 4: 218–219. doi: 10.1016/j.ijpddr.2014.10.001 PMID: 25516831
- Matthews J. Mitigating anthelmintic resistance in equine nematodes. Int J Parasitol Drugs Drug Resist. 2014; 4: 310–315. doi: <u>10.1016/j.ijpddr.2014.10.003</u> PMID: <u>25516842</u>
- Wang W, Wang L, Liang YS. Suceptibility or resistance of praziquantel in human schistosomiasis: a review. Parasitol Res. 2012; 111: 1871–1877. doi: 10.1007/s00436-012-3151-z PMID: 23052781
- Mwangi IN, Sanchez MC, Mkoji GM, Agola LE, Runo SM, Cupit PM, Cunningham C. Praziquantel sensitivity of Kenyan Schistosoma mansoni isolates and the generation of a laboratory strain with reduced susceptibility to the drug. Int J Parasitol Drugs Drug Resist. 2014; 4: 296–300. doi: <u>10.1016/j.ijpddr.</u> 2014.09.006 PMID: <u>25516840</u>
- Stothard JR, Sousa-Figueiredo JC, Betson M, Bustinduy A, Reinhard-Rupp J. Schistosomiasis in African infants and preschool children: let them now be treated! Trends Parasitol. 2013; 29: 197–205. doi: 10.1016/j.pt.2013.02.001 PMID: 23465781
- Harvey AL, Edrada-Ebel R, Quinn RJ. The re-emergence of natural products for drug discovery in the genomics era. Nat Rev Drug Discov. 2015; 2:111–29.
- 14. Lai H, Sasaki T, Singh NP. Targeted treatment of cancer with artemisinin and artemisinin tagged ironcarrying compounds. Expert Opin Ther Targets. 2005; 9: 995–1007. PMID: <u>16185154</u>
- Athanasiadou S, Githiori J, Kyriazakis I. Medicinal plants for helminth parasite control: facts and fiction. Animal. 2007; 1: 1392–1400. doi: <u>10.1017/S1751731107000730</u> PMID: <u>22444894</u>
- Wangchuk P, Keller PA, Pyne SG, Taweechotipatr M, Tonsomboon A, Rattanajak R, Kamchonwongpaisan S. Evaluation of an ethnopharmacologically selected Bhutanese medicinal plants for their major classes of phytochemicals and biological activities. J Ethnopharmacol. 2011; 137: 730–742. doi: <u>10.</u> <u>1016/j.jep.2011.06.032</u> PMID: <u>21741462</u>
- Wangchuk P, Keller PA, Pyne SG, Willis AC, Kamchonwongpaisan S. Antimalarial alkaloids from a Bhutanese traditional medicinal plant *Corydalis dubia*. J Ethnopharmacol. 2012; 143: 310–313. doi: <u>10.1016/j.jep.2012.06.037</u> PMID: <u>22796506</u>
- Wangchuk P, Keller PA, Pyne SG, Lie W, Willis AC, Rattanajak R, Kamchonwongpaisan S. A new protoberberine alkaloid from *Meconopsis simplicifolia* (D. Don) Walpers with potent antimalarial activity against a multidrug resistant Plasmodium falciparum strain. J Ethnopharmacol. 2013a; 150: 953–959.
- Wangchuk P, Pyne SG, Keller PA, Taweechotipatr M, Kamchonwongpaisan S. Phenylpropanoids and furanocoumarins as antibacterial and antimalarial constituents of the Bhutanese medicinal plant *Pleur*ospermum amabile. Nat Prod Commun. 2014; 9: 957–960. PMID: 25230503
- Edwards J, Brown M, Peak E, Bartholomew B, Nash RJ, Hoffman KF. The diterpenoid 7-Keto-sempervirol, derived from Lycium chinense, displays anthelmintic activity against both Schistosoma mansoni and Fasciola hepatica. PLoS Negl Trop Dis. 2015; doi: 10.1371/journal.pntd.0003604
- Wangchuk P, Keller PA, Pyne SG, Korth J, Samten Taweechotipatr M, Rattanajak R, Kamchonwongpaisan S. Antimicrobial, antimalarial and cytotoxicity activities of constituents of a Bhutanese variety of *Ajania nubigena*. Nat Prod Commun. 2013b; 8: 733–736.
- Ballabh B, Chaurasa OP. Medicinal plants of cold desert Ladakh used in the treatment of stomach disorders. Indian J Tradit Know. 2009; 8: 185–190.
- Pareek A, Suthar M, Rathore GS, Bansal V. Feverfew (*Tanacetum parthenium* L.): A systematic review. Pharmacogn Rev. 2011; 5:103–10. doi: <u>10.4103/0973-7847.79105</u> PMID: <u>22096324</u>

- 24. Smout M, Kotze AC, McCarthy JS, Loukas A. A novel high throughput assay for anthelmintic drug screening and resistance diagnosis by real-time monitoring of parasite motility. PLOS Negl Trop Dis. 2010; 4: e885. doi: 10.1371/journal.pntd.0000885 PMID: 21103363
- Rinaldi G, Loukas A, Brindley PJ, Irelan PJ, Smout MJ. Viability of developmental stages of Schistosoma mansoni quantified with xCELLigence worm real-time motility assay (xWORM). Int J Parasitol Drugs Drug Resist. 2015; 5: 141–148. doi: 10.1016/j.ijpddr.2015.07.002 PMID: 26288742
- Harn DA, Mitsuyama M, David JR. Schistosoma mansoni: anti-egg monoclonal antibody protects against cercarial challenge in vivo. J Exp Med. 1984; 159: 1371–1387. PMID: 6538899
- Basch PF. Cultivation of Schistosoma mansoni in vitro. I. Establishment of cultures from cercariae and development until pairing. J Parasitol. 1981; 67: 179–185. PMID: <u>7241277</u>
- Peak E, Chalmers IW, Hoffmann KF. Development and validation of a quantitative, high-throughput, fluorescent-based bioassay to detect *Schistosoma* viability. PloS Negl Trop Dis. 2010; 4: e759. doi: <u>10.</u> <u>1371/journal.pntd.0000759</u> PMID: <u>20668553</u>
- Rajasekariah GR, Deb BN, Jones MP, Dhage KR, Bose S. Response of pre-adult and adult stages of *Trichuris muris* to common anthelmintics in mice. Int J Parasitol. 1991; 21: 697–702. PMID: <u>1757197</u>
- Stepek G, Lowe AE, Buttle DJ, Duce IR, Behnke JM. *In vitro* and *in vivo* anthelmintic efficacy of plant cysteine proteinases against the rodent gastrointestinal nematode, *Trichuris muris*. Parasitology. 2006; 132: 681–689. PMID: 16448585
- Tritten L, Silbereisen A, Keiser J. Nitazoxanide: *In vitro* and *in vivo* drug effects against *Trichuris muris* and *Ancyclostoma ceylanicum*, alone or in combination. Int J Parasitol Drugs Drug Resist. 2012; 2: 98–105. doi: 10.1016/j.ijpddr.2012.02.004 PMID: 24533270
- De Oliveira CNF, de Oliveira RN, Frezza TF, Rehder VLG, Allergretti SM. Tegument of Schistosoma mansoni as a therapeutic target. In: INTECH open science. 2013. pp.152–177. Available: <u>http://dx.doi.org/10.5772/53653</u>.
- Becker B, Mehlhom H, Andrews P, Thomas H, Eckert J. Light and electron microscopic studies on the effect of praziquantel on Schistosoma mansoni, Dicrocoelium dendriticum, and Fasciola hepatica (Trematoda) in vitro. Z Parasitenkd. 1980; 63: 113–128. PMID: <u>7456640</u>
- Kohn A, Lopez-Alvarez ML, Katz N. Transmission and scanning electron microscopical studies in the tegument of male Schistosoma mansoni after oxamniquine treatment. Ann Parasitol Hum Comp. 1982; 57: 285–291. PMID: 7137803
- Shaw MK, Erasmus DA. Schistosoma mansoni: changes in elemental composition in relation to the age and sexual status of the worms. Parasitology. 1983; 86: 439–453. PMID: <u>6877870</u>
- Brickes CS, Depenbusch JW, Benett JL, Thompson DF. The relationship between tegumental disruption and muscle contraction in *Schistosoma mansoni* exposed to various compounds. Parasitol Res. 1983; 69: 61–67.
- Manneck T, Haggenmuller Y, Keiser J. Morphological effects and tegumental alterations induced by mefloquine on schistosomula and adult flukes of Schistosomula and adult flukes of Schistosoma mansoni. Parasitology. 2010; 137: 85–98. doi: 10.1017/S0031182009990965 PMID: 19814844
- Abdul-Ghani R, Loutfy N, Sheta M, Hassan A. Artemether shows promising female schistosomacidal and ovicidal effects on the Egyptian strain of *Schistosoma mansoni* after maturity of infection. Parasitol Res. 2011; 108: 1199–1205. doi: 10.1007/s00436-010-2163-9 PMID: 21107863
- Zhang SM, Coultas KA. Identification of plumbagin and sanguinarine as effective chemotherapeutic agents for treatment of schistosomiasis. Int J Parasitol Drugs Drug Resist. 2013; 3: 28–34. PMID: 23641325
- Matadamas-Martínez F, Nogueda-Torres B, Hernández-Campos A, Hernández-Luis F, Castillo R, Mendoza G, Ambrosio JR, Andrés-Antonio G, Yépez-Mulia L. Analysis of the effect of a 2-(trifluoromethyl)-1H-benzimidazole derivative on *Trichinella spiralis* muscle larvae. Vet Parasitol. 2013; 194: 193–197. doi: 10.1016/j.vetpar.2013.01.054 PMID: 23462252
- Hanser E, Mehlhorn H, Hoeben D, Vlaminck K. In vitro studies on the effects of flubendazole against Toxocara canis and Ascaris suum. Parasitol Res. 2003; 89: 63–74. PMID: <u>12474046</u>
- Wright KA, Chan J. Sense receptors in the bacillary band of trichuroid nematodes. Tissue Cell. 1973; 5: 373–80. PMID: <u>4744677</u>
- Lee TDG, Wright KA. The morphology of the attachment and probable feeding site of the nematode Trichuris muris (Schrank, 1788) Hall, 1916. Can J Zool. 1978; 56: 1889–1905. PMID: 737601
- Huttemann M, Schmahl G, Mehlhom H. Light and electron microscopic studies on two nematodes, *Angiostrongylus cantonensis* and *Trichuris muris*, differing in their mode of nutrition. Parasitol Res. 2007; 101: S225–S232. doi: 10.1007/s00436-007-0698-1 PMID: 17823832

- Torres EJL, de Souza W, Miranda K. Comparative analysis of *Trichuris muris* surface using conventional, low vacuum, environmental and field emission scanning electron microscopy. Vet Parasitol. 2013; 196: 409–416. doi: <u>10.1016/j.vetpar.2013.02.026</u> PMID: <u>23537947</u>
- Lipinski CA, Lombardo F, Dominy BW, Feeney PJ. Experimental and computational approaches to estimate solubility and permeability in drug discovery and development settings. Adv Drug Deliv Rev. 2001; 46: 3–26. PMID: <u>11259830</u>